Author's response to reviews

Title: Interaction of a traditional Chinese Medicine (PHY906) and CPT-11 on the inflammatory process in the tumor microenvironment

Authors:

Ena Wang (ewang@cc.nih.gov)
Scott Bussom (sbussomyale@gmail.com)
Jinguo Chen (Jinguo.Chen@nih.gov)
Courtney Quinn (courtneyaq@gmail.com)
Davide Bedognetti (Davide.Bedognetti@nih.gov)
Wing Lam (lamwingyale@hotmail.com)
Fulan Guan (fulan.guan@yale.edu)
Zaoli Jiang (jiang.zaoli@gmail.com)
Yichao Mark (ymark1220@gmail.com)
Yingdong Zhao (Yingdong.Zhao@nih.gov)
David F Stroncek (David.Stroncek@nih.gov)
Jeffrey White (Jeffrey.White@nih.gov)
Francesco M Marincola (FMarincola@mail.cc.nih.gov)
Yung-Chi Cheng (yccheng@yale.edu)

Version: 6 Date: 6 May 2011

Author's response to reviews: see over
May 5, 2011

Dear Editor of BMC Medical Genomics:
Thank you for your favorable review of our manuscript MS: 1578451354498788
Interaction of a traditional Chinese Medicine (PHY906) and CPT-11 on the inflammatory process in the tumor microenvironment.
We thank editor for their positive and constructive remarks. We addressed the requests. We hope you will find the manuscript acceptable for publication in the present format.

Major revisions (we require the author to make these changes)

Additional files: please make sure you cite the additional file when referring to tables or figures in them.
The changes were made on Pages 11 and 12

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Structure: please change your 'Introduction' to 'Background'. Please move your 'Acknowledgements' so that they appear after the 'Author Contributions'.
We performed the requested changes.

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
We removed all highlighting and tracking.

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
We performed the requested changes except for the ref n 12 (article in press).

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Thanks for your checking

Thanks
Thank you again for your consideration
Sincerely,

[Signature]